Ther Adv Gastroenterol

2025, Vol. 18: 1–25

17562848251329145

© The Author(s), 2025. Article reuse guidelines: sagepub.com/journalspermissions

Correspondence to: Enrique Rodríguez de Santiago

Department of Gastroenterology and Hepatology, Hospital Universitario Ramón y Cajal, Instituto Ramón y Cajal de Investigación Sanitaria (IRYCIS), Madrid, Spain

Centro de Investigación Biomédica en Red en el Área temática de Enfermedades Hepáticas y Digestivas (CIBEREHD), M-607, Carretera de Colmenar Viejo Km 9,100. 28034 Madrid, Spain **enrodesan(Agmail.com** 

#### Sandra Pérez de la Iglesia Marta Aicart-Ramos Miguel Ángel Rodríguez Gandía

Department of Gastroenterology and Hepatology, Hospital Universitario Ramón y Cajal, Instituto Ramón y Cajal de Investigación Sanitaria (IRYCIS), Madrid, Spain

### Diego de Frutos

Department of Gastroenterology, Hospital Universitario Puerta de Hierro, Instituto de Investigación Sanitaria Puerta de Hierro-Segovia de Arana, Majadahonda, Spain

#### José Carlos Marín-Gabriel María López-Cerón Pinilla

Endoscopy Unit, Department of Gastroenterology, i+12 Research Institute, Hospital Universitario 12 de Octubre, Universidad Complutense de Madrid, Madrid, Spain

Carolina Mangas-SanJuan Endoscopy Unit, Hospital General Universitario Dr. Balmis, Instituto de Investigación Sanitaria y Biomédica de Alicante (ISABIAL), Alicante, Spain

# Delphi consensus statement for the management of delayed post-polypectomy bleeding

Enrique Rodríguez de Santiago\*<sup>(D)</sup>, Sandra Pérez de la Iglesia\*<sup>(D)</sup>, Diego de Frutos, José Carlos Marín-Gabriel<sup>(D)</sup>, Carolina Mangas-SanJuan<sup>(D)</sup>, Raúl Honrubia López, Hugo Uchima<sup>(D)</sup>, Marta Aicart-Ramos<sup>(D)</sup>, Miguel Ángel Rodríguez Gandía, Eduardo Valdivielso Cortázar<sup>(D)</sup>, Felipe Ramos Zabala, Marco Antonio Álvarez, Marina Solano Sánchez<sup>(D)</sup>, Jesús Manuel González Santiago, Eduardo Albéniz<sup>(D)</sup>, Gonzalo Hijos-Mallada<sup>(D)</sup>, Nerea Castro Quismondo, Miguel Fraile-López, David Martínez Ares, Javier Tejedor-Tejada<sup>(D)</sup>, Luis Hernández, Joan B. Gornals<sup>(D)</sup>, Sergi Quintana-Carbo<sup>(D)</sup>, Juan Ocaña, João A. Cunha Neves<sup>(D)</sup>, Juan Martínez Martínez, María López-Cerón Pinilla, Carlos Dolz Abadía and María Pellisé<sup>(D)</sup>; on behalf of the Mucosal Resection and Third-Space Endoscopy Working Group from the Spanish Society of Gastrointestinal Endoscopy

## Abstract

**Background:** Delayed post-polypectomy bleeding (DPPB) is the most common adverse event following colonic polypectomy, yet its management remains highly heterogeneous and lacks standardization. A considerable number of colonoscopies performed for DPPB may be unnecessary and do not result in hemostatic intervention.

**Objectives:** To develop evidence-based statements to guide clinical decision-making in DPPB. **Design:** Multidisciplinary Delphi consensus statement.

**Methods:** A panel of 29 experts in gastroenterology, hematology, radiology, and surgery was assembled. Through a systematic review of the literature and a modified Delphi process, consensus statements were developed through iterative rounds of anonymous voting. Statements were revised following anonymous voting and feedback at each round. Those achieving 80% agreement were accepted.

**Results:** The expert panel reached a consensus on 36 statements, covering areas such as antithrombotic management, bowel preparation, colonoscopy indications, and therapeutic hemostatic modalities. Key recommendations include guidance for managing self-limited bleeding and risk stratification to reduce the rate of unnecessary colonoscopies, as well as recommendations for hemodynamically unstable patients who may require primary angioembolization. A practical clinical algorithm is proposed.

**Conclusion:** This document provides a consensus-based framework for managing DPPB. These recommendations aim to improve patient outcomes and optimize healthcare resources while fostering a standardized approach to this common adverse event.

# Plain language summary

# Expert consensus on the management of post-polypectomy bleeding after colonoscopy

Delayed post-polypectomy bleeding (DPPB) is the most common complication following the removal of polyps during a colonoscopy. Despite its frequency, there is no standardized approach for managing this issue. Many patients undergo colonoscopies that may not be necessary and do not lead to further treatment. This can result in wasted healthcare resources and unnecessary patient discomfort. To address this, a group of 29 experts in

journals.sagepub.com/home/tag



Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-NonCommercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the Sage and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).

## THERAPEUTIC ADVANCES in Gastroenterology

#### Raúl Honrubia López

Department of Gastroenterology, Hospital Universitario Infanta Sofía, Faculty of Biomedical and Health Sciences, Universidad Europea de Madrid, San Sebastián de los Reyes, Madrid, Spain

### Hugo Uchima

Department of Gastroenterology, Hospital Universitario Germans Trias i Pujol, Badalona, Spain

Centro Médico Teknon, Barcelona, Spain

#### Eduardo Valdivielso Cortázar

Department of Gastroenterology, Complejo Hospitalario Universitario de A Coruña, A Coruña, Spain

#### Felipe Ramos Zabala

Department of Gastroenterology, Hospital Universitario HM Montepríncipe, HM Hospitales, Madrid, Spain Universidad San Pablo-

CEU, CEU Universities, Madrid, Spain

#### Marco Antonio Álvarez Gastrointestinal Oncology Endoscopy and Surgery (GOES) Research Group, Althaia Xarxa Assistencial Universitària de Manresa, Institut de Recerca i Innovació en Ciències de la Vida i de la Salut de la Catalunya Central (IRIS-CC), Manresa, Barcelona, Spain

## Marina Solano Sánchez

Department of Gastroenterology, Hospital Comarcal de Alcañiz, Grupo de investigación INDOGASTRO, Zaragoza, Spain

#### Jesús Manuel González Santiago

Department of Gastroenterology and Hepatology, University Hospital of Salamanca, Institute for Biomedical Research of Salamanca (IBSAL), Salamanca, Spain

### Eduardo Albéniz

Endoscopy Unit, Department of Gastroenterology, Hospital Universitario de Navarra Navarrabiomed Biomedical Research Center, Universidad Pública de Navarra (UPNA), Instituto de Investigación Sanitaria de Navarra (IdiSNA), Pamplona, Spain

#### **Gonzalo Hijos-Mallada** Department of Gastroenterology, Hospital Clínico Universitario

gastroenterology, hematology, radiology, and surgery reviewed the available scientific evidence. Using a structured process called the Delphi method, they developed 36 consensus statements to guide healthcare professionals. These statements provide clear recommendations for managing DPPB, such as when to perform a colonoscopy, how to handle antithrombotic medications (like blood thinners), and what treatments to use for severe bleeding. One key takeaway is that many cases of DPPB resolve on their own without the need for a repeat colonoscopy. For more severe cases, the guidance outlines the use of endoscopic treatments, radiological or surgical interventions if necessary. A practical algorithm is included to help doctors make decisions tailored to individual patients. These recommendations aim to improve patient outcomes, reduce unnecessary procedures, and optimize the use of healthcare resources. By fostering a consistent approach to managing DPPB, this consensus provides a valuable tool for healthcare professionals worldwide.

Keywords: colonic polyps, colonoscopy, hemorrhage, polypectomy

Received: 2 January 2025; revised manuscript accepted: 6 March 2025.

## Introduction

Colonic polypectomy is a critical intervention for preventing colorectal cancer and has become the most common endoscopic procedure.<sup>1</sup> Delayed post-polypectomy bleeding (DPPB) is the most frequent adverse event (AE) of this technique.<sup>2</sup> Existing research primarily concentrates on assessing the incidence, risk factors, and efficacy of hemoclipping and other preventive methods.<sup>2–5</sup> However, there is a scarcity of studies exploring the management of DPPB.

Available data indicate that DPPB is often selflimited, suggesting an overutilization of colonoscopy.<sup>6–8</sup> The decision-making process in these cases typically relies on the discretion of the treating physician or local clinical protocols. This accounts for the clinical management variability observed across different hospitals and the frequent occurrence of colonoscopies without clear benefit, ranging from 0% to 63%.<sup>2,6–12</sup> Although repeating a colonoscopy may warrant endoscopic therapy in certain patients, many could benefit from a watchful waiting approach, as spontaneous resolution is relatively common.<sup>6–8</sup>

Current clinical guidelines from international scientific societies lack a comprehensive algorithm detailing when and for whom a colonoscopy should be performed. Reducing the number of ineffective colonoscopies would result in decreased costs, less consumption of healthcare resources, and avoidance of potential AEs associated with the procedure. Moreover, reducing colonoscopy frequency could improve the patient experience by avoiding the discomfort associated with the preparation and the procedure itself. In addition, the environmental impact of colonoscopy, which can reach up to 30 kg of  $CO_2$  per procedure, should not be underestimated.<sup>13</sup>

An analysis of the literature highlights the necessity to generate clinical recommendations that optimize and standardize the clinical management of DPPB. Consequently, the mucosal resection and third-space group of the Spanish Society of Gastrointestinal Endoscopy (SEED) proposed the creation of a consensus statement to address this gap. This document aims to conduct a critical appraisal of the available evidence to generate a management algorithm based on the best available evidence for DPPB.

### Methods

The project was promoted by the steering committee of the SEED group on mucosal resection and third-space endoscopy. The expert panel consisted of consultant gastroenterologists with a special focus on gastrointestinal endoscopy or clinical inpatient management, as well as hematologists, radiologists, and surgeons. The primary aim was to reach a consensus on a clinical management algorithm for delayed DPPB, regardless of the polyp removal technique used. The scope of the document did not include the prophylaxis of DPPB, given that contemporary clinical guidelines were readily available.<sup>14</sup>

A Consensus Statement format was deemed appropriate given the expected limited body of evidence. In April 2023, the steering committee outlined the manuscript's structure, established Task forces, and developed questions of interest in the PICO format (Population/Intervention/ Comparisons/Outcomes). Systematic literature reviews were conducted in at least two databases. Task forces critically assessed the literature using the Grading of Recommendations Assessment, Development, and Evaluation (GRADE) system through May 2024. In the absence of evidence or when GRADE was not applicable, recommendations were formulated as Good Practice Statements<sup>15</sup> or statements. Consensus among members was determined anonymously using a Delphi method. Statements were rated using a 5-point Likert scale on a web-based platform (Google Forms), with consensus defined as ≥80% agreement (combining Agree and Strongly agree). The final manuscript received the SEED governing board's endorsement. The manuscript is reported following the CREDES (Conducting and REporting of DElphi Studies) guideline.<sup>16</sup> Additional methodology information is provided in Supplemental Material.

## Definitions

### Statement 1

Delayed post-polypectomy bleeding is defined as any rectal bleeding occurring within 30 days after the completion of the index colonoscopy, requiring emergency room presentation, hospitalization, or reintervention (repeat endoscopy, angiography, or surgery). (Strongly agree: 69.2%; agree: 30.8%)

### Statement 2

Stigmata of recent bleeding is defined as the presence of any of the following findings at a polyp resection mucosal defect: active bleeding (either jet or oozing), a visible vessel, or an adherent clot that remains despite targeted jet irrigation.

(Strongly agree: 71.4%; agree: 25%)

### Statement 3

Rebleeding is defined as (1) a new episode of hematochezia or rectorrhagia at least 48 hours after successful hemostatic therapy of the initial delayed post-polypectomy bleeding or after discharge; (2) a new drop in hemoglobin levels greater than 3g/dL or additional transfusion requirements 48 hours after resolution of the baseline bleeding episode; (3) confirmation of stigmata of recent bleeding at the scar responsible for the initial delayed post-polypectomy bleeding on repeat endoscopy or angio-CT/angiography; and (4) any colonoscopy performed for suspicion of bleeding in research studies.

(Strongly agree: 50%; agree: 42.3%)

The definitions provided in this document are based on the available literature (see Supplemental Table 1) and agreed-upon consensus, prioritizing clinical relevance and applicability. They are intended to be used both in clinical practice and for research purposes in future studies.

### Epidemiology

The incidence of DPPB ranges from 0.2% to 7%, depending on the type of polyp, resection technique used, or the presence of risk factors such as active use of antithrombotics.<sup>2–4,9–11,17</sup> Although specific data are scarce, reported transfusion rates vary widely from 5.5% to 75%, with a median hospitalization length of 3 days and a 30-day mortality rate of 0.18%.<sup>2,8,17–19</sup> No information is currently available on the costs associated with DPPB. The rebleeding rate ranges from 7.6% to 14.9%, although predictive factors are still not well established.<sup>8–10,12</sup> Details of the individual studies are provided in Supplemental Table 1.

### Prognostic scales and grading severity Statement 4

SEED recommends that clinical judgment is crucial in assessing patients with delayed post-polypectomy bleeding, with prognostic scales or tests serving as complementary tools rather than substitutes.

Good Practice Statement. (Strongly agree: 84.6%; agree: 15.4%)

In recent years, several prognostic scales have been developed for lower gastrointestinal bleeding (LGIB).<sup>20–24</sup> Most studies are observational, aiming to identify key factors associated with LGIB prognosis. However, comprehensive validation of these prognostic indices remains incomplete, and none of them is primarily focused on DPPB.<sup>20–24</sup>

Among the various scales, the Oakland score has emerged as one of the most extensively studied. A recent meta-analysis highlighted its superior prognostic performance in predicting outcomes such as safe discharge, likelihood of major bleeding, and need for transfusion.<sup>25</sup> Importantly, the Lozano Blesa, Instituto de Investigación Sanitaria (IIS), Aragón, Zaragoza, Spain

### Nerea Castro Quismondo Department of Hematology

and Hemotherapy, Hospital Universitario 12 de Octubre, Madrid, Spain

### Miguel Fraile-López

Department of Gastroenterology and Hepatology, Clinical and Translational Research in Digestive Diseases, Valdecilla Research Institute (IDIVAL), Marqués de Valdecilla University Hospital, Santander, Spain

### David Martínez Ares

Department of Gastroenterology— Ingaled, Hospital Quirónsalud Miguel Domínguez, Pontevedra, Spain

### Department of

Gastroenterology, Hospital Quirónsalud A Coruña, A Coruña, Spain

### Javier Tejedor-Tejada Department of

Gastroenterology, Hospital Universitario de Cabueñes, Gijón, Spain

### Luis Hernández

Department of Gastroenterology, Hospital Santos Reyes, Aranda de Duero, Spain

### Joan B. Gornals

Sergi Quintana-Carbo Department of Digestive Diseases, Hospital Universitari de Bellvitge, Universitat de Barcelona, Bellvitge Biomedical Research Institute (IDIBELL), Barcelona, Spain

### Juan Ocaña

Department of Surgery, Colorectal Surgery Unit, Hospital Universitario Ramón y Cajal, Madrid, Spain

### João A. Cunha Neves

Department of Gastroenterology, Unidade Local de Saúde do Algarve, Hospital de Portimão, Portimão, Portugal

### Juan Martínez Martínez

Department of Radiology, Hospital General Universitario Dr. Balmis, Instituto de Investigación Sanitaria y Biomédica de Alicante (ISABIAL), Alicante, Spain

### Carlos Dolz Abadía

Department of Gastroenterology, Hospital Juaneda Miramar, Palma de Mallorca, Spain

# THERAPEUTIC ADVANCES in Gastroenterology

### María Pellisé

Centro de Investigación Biomédica en Red en el Área temática de Enfermedades Hepáticas y Digestivas (CIBEREHD), Madrid, Spain

Department of Gastroenterology, Hospital Clinic Barcelona, Institut d'Investigacions Biomediques August Pi i Sunyer (IDIBAPS), Barcelona, Spain

Facultat de Medicina i Ciencies de la Salud, Universitat de Barcelona, Barcelona, Spain

\*Joint first co-authorship.

Oakland score has been particularly noted for its ability to identify patients suitable for early discharge with minimal risk of rebleeding. A score of 8 or lower in the Oakland score correlates with a 95% probability of safe discharge without additional complications,<sup>21</sup> and it has been cited in recent guidelines as a valid cutoff for clinical practice.<sup>26</sup> Similarly, the meta-analysis identified the Strate score as the most effective predictor of the need for hemostasis.<sup>25</sup>

Despite the array of available scoring systems, current evidence does not support the exclusive endorsement of any single scale.<sup>26–28</sup> The Oakland score, validated primarily in stable patients without active bleeding or anticoagulant therapy, appears most suited for predicting early discharge. However, its accuracy in patients presenting with DPPB remains unknown and may differ from other LGIB etiologies.

### Statement 5

SEED suggests utilizing an endoscopy-focused classification (AGREE classification or ASGE Lexicon) for grading the severity of delayed post-polypectomy bleeding.

Weak recommendation, very low-quality evidence. (Strongly agree: 53.8%; agree: 34.6%)

The use of a standardized classification for grading the severity of AEs in endoscopy is essential for assessing procedure safety, allowing for direct comparisons, and represents a key element of high-quality colonoscopy. However, only a few classifications have been proposed, and none specifically for DPPB. These include the Clavien-Dindo Classification, developed in the surgical field in 2004,29 the American Society for Gastrointestinal Endoscopy (ASGE) lexicon proposed in 2010,30 and the AGREE (Adverse events in GastRointEstinal Endoscopy) classification introduced in 2021.<sup>31</sup> The AGREE classification and ASGE lexicon may be more appropriate for grading severity of DPPB, as they are specifically designed for endoscopic procedures and have been utilized in previous DPPB studies.7,8,32,33 We did not find any studies comparing the different classifications in the context of DPPB.

# Initial management and blood transfusion *Statement 6*

SEED recommends evaluating the patient's general clinical condition and prioritizing resuscitation

measures if necessary. The initial assessment should include a focused medical history, a physical examination (including an anorectal exam), and blood tests.

Good Practice Statement. (Strongly agree: 92.3%; agree: 7.7%)

There are no studies evaluating initial supportive measures in patients with DPPB. Recommendations are derived from expert consensus and clinical guidelines aimed at managing LGIB. Initially, the patient's hemodynamic stability should be assessed, prioritizing resuscitation measures.<sup>26,34</sup> It is recommended to secure two adequately sized peripheral veins and/or a central line.<sup>35</sup> Crystalloids are usually preferred over colloids for volume replacement and stabilization.36 The initial patient evaluation should involve a focused medical history that details colonoscopy information (date, lesions removed, size, location, removal technique, symptom onset, and prophylactic scar therapy), relevant comorbidities, current pharmacological treatments (such as antiplatelets and anticoagulants), a rectal examination, and blood tests including a complete blood count and coagulation studies.<sup>26,34,37</sup>

There are no studies comparing out versus inpatient management in patients with DPPB. Most studies do not specify the criteria for patient admission or discharge. In a retrospective study of 548 patients with DPPB, 80.8% were admitted, but admission criteria were not detailed.<sup>8</sup> Another study of 69 DPPB patients found that 69.6% underwent therapeutic colonoscopy as inpatients, while the rest were managed as outpatients, with no details on observation time or discharge criteria.<sup>10</sup>

Other authors advocate for admitting all patients.<sup>6,38</sup> Burgess et al. propose an algorithm recommending 24h of surveillance and considering discharge if there is no recurrence of bleeding, the patient remains hemodynamically stable, and there is no significant change in hemoglobin level during that time.<sup>7</sup> Van der Star et al. evaluated 42 patients and suggested discharge if bleeding stops spontaneously and does not recur during 24h of observation.<sup>9</sup>

## Statement 7

In hemodynamically stable patients with significant cardiovascular comorbidity, red blood cell transfusion is recommended for hemoglobin levels  $\leq 8-9g/dL$ with a post-transfusion target of 9-10g/dL. In hemodynamically stable patients without cardiovascular comorbidity, the hemoglobin threshold for blood transfusion should be  $\leq 7 g/dL$ , with a target range of 7–9 g/dL.

Strong recommendation, very low-quality evidence. (Strongly agree: 69.2%; agree: 30.8%)

### Statement 8

SEED recommends following severe bleeding protocols for blood transfusion in hemodynamically unstable patients, rather than relying solely on hemoglobin levels.

Good Practice Statement. (Strongly agree: 60.7%; agree: 39.3%)

No specific recommendations regarding transfusion for DPPB have been identified. A post hoc study in patients with LGIB<sup>39</sup> concluded that, although there were no statistically significant differences between restrictive strategies (Hb threshold <7 mg/dL) and liberal strategies (transfusion threshold <9 mg/dL), a trend toward reduced rebleeding and severe events was observed with restrictive strategies. The recommendation for restrictive transfusion strategies is primarily extrapolated from evidence in upper gastrointestinal bleeding.<sup>40</sup> Patients with significant cardiovascular comorbidity or advanced age may benefit from a less restrictive transfusion threshold. Given the lack of specific studies on DPPB, our recommendations are in line with current guidelines on LGIB.26,34,37

For patients with hemorrhagic shock, a different approach is warranted, involving volume resuscitation and blood component transfusion according to severe bleeding protocols. Platelet transfusion and fibrinogen replacement should be guided by laboratory parameters.<sup>41–44</sup>

### **Management of anti-thrombotics**

The SEED proposal for antithrombotic management is summarized in Figures 1 and 2. Our recommendations for the management of antithrombotics have been adapted and adhere to current LGIB guidelines, given the lack of specific evidence for DPPB.<sup>26,34,37</sup>

## Antiplatelet discontinuation

### Statement 9

SEED recommends that aspirin for secondary prophylaxis should not be withheld. It may be temporarily withheld only in cases of severe ongoing bleeding and/ or hemodynamic instability. Strong recommendation, very low-quality evidence. (Strongly agree: 61.5%; agree: 38.5%)

### Statement 10

SEED recommends withholding aspirin for primary prophylaxis during the acute phase of DPPB. Strong recommendation, very low-quality evidence. (Strongly agree: 61.5%; agree: 38.5%)

### Statement 11

SEED recommends against discontinuing dual-antiplatelet therapy (aspirin and a P2Y12 receptor antagonist) in patients with delayed post-polypectomy bleeding. In cases of severe ongoing bleeding and/or hemodynamic instability, discontinuing the P2Y12 receptor antagonist while continuing aspirin may be considered. This approach should not be undertaken without prior consultation with a cardiologist.

Strong recommendation, very low-quality evidence. (Strongly agree: 61.5%; agree: 34.6%)

In 2021, the European Society of Gastrointestinal Endoscopy (ESGE) recommended withholding low-dose aspirin (LDA) during LGIB if taken for primary prophylaxis. However, the maintenance of LDA was advocated in secondary prevention indications. Routine discontinuation of dual-antiplatelet therapy (DAPT) was not recommended. LDA should be maintained, and the P2Y12 receptor antagonist can be either continued or temporarily withheld, depending on the severity of bleeding and ischemic risk.<sup>26</sup>

Evidence regarding the management of antiplatelet therapy in LGIB is limited, and evidence is even more limited in the context of DPPB, with no specific studies conducted to date. A cohort study reported a higher 1-year all-cause mortality (hazard ratio 2.16, 95% confidence interval (CI): 1.39-3.35) among 548 patients taking LDA for secondary prevention if it was interrupted following an episode of gastrointestinal bleeding (16% LGIB).45 This result was in line with previous observational data.46 While continuing antiplatelets for secondary prophylaxis may be associated with a relatively higher risk of rebleeding,<sup>46–49</sup> this was not consistently detected in all studies.37,50,51 The potential beneficial impact on ischemic risk<sup>46,47,49</sup> and mortality<sup>45-47</sup> suggests that continuation is the preferred approach. Therefore, we recommend refraining from routinely withholding LDA when indicated for secondary prevention.



**Figure 1.** Delayed post-polypectomy bleeding in patients on anticoagulants. DOAC, direct oral anticoagulants; PCC, prothrombin complex concentrate; VKA, vitamin K antagonist.

This recommendation does not apply to LDA for primary prevention, as current guidelines restrict its use to a very limited population and discourage its use in those with an increased bleeding risk.<sup>52</sup> Consequently, we recommend withholding LDA in primary prevention patients



**Figure 2.** Delayed post-polypectomy bleeding in patients on antiplatelets. DAPT, dual-antiplatelet therapy; LDA, low-dose aspirin.

presenting with DPPB and reassessing its indication.

Evidence regarding the management of DPPB patients on DAPT (LDA and a P2Y12 receptor antagonist) is limited. Most of these patients have undergone coronary intervention and stent placement within the last 6-12 months, leading to a significantly high ischemic risk. A multidisciplinary approach involving cardiology is essential, as discontinuing DAPT after acute coronary syndrome and stent placement is linked to an increased risk of ischemic events and mortality.53 Cardiology guidelines currently accept a shorter period of DAPT (from 1 to 3-6 months) in patients with high bleeding risk. Therefore, the timing of the DPPB episode relative to the prior stent placement is crucial in determining whether to withdraw the P2Y12 receptor antagonist.54 Data from a randomized controlled trial (RCT), conducted in a limited number of patients with

upper gastrointestinal bleeding, suggest that continuation of LDA in DAPT patients is associated with a reduced incidence of ischemic events in the short term.<sup>55</sup> Therefore, since most DPPB cases are self-limited and have lower mortality and rebleeding rates compared to other LGIB etiologies,<sup>8</sup> we recommend against the routine discontinuation of DAPT, particularly within the first 1–3 months following acute coronary syndrome. In severe cases with associated hemodynamic instability, temporary interruption of the P2Y12 receptor antagonist while maintaining LDA may be considered.<sup>26,34,37</sup>

### Antiplatelet resumption Statement 12

SEED suggests that low-dose aspirin and P2Y12 receptor antagonists should be resumed within 5 days or earlier in the absence of bleeding or if hemostasis is achieved.

No data exist on the optimal time to resume antiplatelet agents in the setting of DPPB. A multicenter prospective analysis on the short-term outcomes of withholding antithrombotic therapy revealed higher in-hospital rebleeding rates in LGIB patients on antiplatelet agents.47 While most rebleeding episodes occurred within 5 days of admission, they were comparable to cases where therapy was uninterrupted or resumed within 5 days. Also, a retrospective cohort study (162 with LGIB) found no significant difference in rebleeding rates when antiplatelet therapy was resumed within 7 days.<sup>56</sup> Another study demonstrated that restarting antiplatelets within 7 days significantly reduced the risk of major adverse cardiovascular and cerebrovascular events, with a higher net clinical benefit, and no significant increase in rebleeding risk.49 Although not specifically for LGIB, this study defined a cutoff of 9 days to resume therapy for net clinical benefit in patients with gastrointestinal bleeding related to P2Y12 receptor antagonists or aspirin therapy.

Interrupting P2Y12 receptor antagonists after coronary stent implantation is consistently linked with a heightened risk of major cardiovascular events.<sup>57,58</sup> Therefore, it is imperative to resume P2Y12 therapy as promptly as feasible following any interruption. Uninterrupted aspirin therapy, with P2Y12 receptor antagonists withheld for no more than 5–7 days, should be reasonable in severe LGIB cases, similar to upper gastrointestinal bleeding.<sup>59</sup>

# Antiplatelet reversal

Statement 13

SEED suggests against platelet transfusions to reverse antiplatelet therapy in patients with delayed post-polypectomy bleeding.

Weak recommendation, very low-quality evidence. (Strongly agree: 61.5%; agree: 30.8%)

Only two relevant studies have been identified from 2013. The first, a case–control study involving patients admitted with gastrointestinal hemorrhage without thrombocytopenia who were on antiplatelet agents showed a significant increase in mortality, along with non-significant increases in recurrent bleeding and thrombotic events associated with platelet transfusion.<sup>60</sup> An RCT involving 190 patients with intracerebral hemorrhage, who were assigned to receive either platelet transfusion or standard care, reported a higher rate of death in the platelet transfusion group (odds ratio (OR) 2.05, 95% CI: 1.18– 3.56).<sup>61</sup> Moreover, this study noted a substantial, albeit not statistically significant, increase in thrombotic events in the platelet transfusion group.<sup>62</sup>

# Anticoagulation withdrawal *Statement 14*

SEED recommends temporarily stopping anticoagulant drugs in cases of delayed post-polypectomy bleeding. However, in selected cases of mild bleeding (Oakland scale score  $\leq 8$ ), continuation of anticoagulation may be considered, provided there are no new signs of bleeding, and the patient remains clinically stable.

Good Practice Statement. (Strongly agree: 53.8%; agree: 38.5%)

None of the reviewed articles offered conclusive evidence on the benefits or harms of discontinuing anticoagulation in the context of DPPB.<sup>56,63–68</sup> The main guidelines offer recommendations on this matter with a weak or very weak level of evidence.<sup>26,34,37,69</sup> In general, they recommend suspending anticoagulation during the bleeding episode but suggest considering its continuation in cases of mild LGIB (Oakland score  $\leq 8$ ). In any case, it seems reasonable to suspend anticoagulation until clinical stability is confirmed and bleeding has ceased.

# Anticoagulation reversal *Statement 15*

SEED suggests reversing anticoagulation only in patients experiencing ongoing severe bleeding and/or hemodynamic instability.

Weak recommendation, low-quality evidence. (Strongly agree: 73.1%; agree: 23.1%)

## Statement 16

In patients with ongoing severe bleeding or hemodynamic instability, SEED recommends withholding vitamin K antagonists and administering intravenous vitamin K and prothrombin complex concentrate (PCC), or fresh frozen plasma if PCC is not available.

Strong recommendation, low-quality evidence. (Strongly agree: 67.9%; agree: 28.6%)

# Statement 17

SEED recommends consulting a local hematologist before using reversal agents in patients on direct oral anticoagulants who are experiencing ongoing delayed post-polypectomy bleeding and hemodynamic instability.

Strong recommendation, low-quality evidence. (Strongly agree: 60.7%; agree: 35.7%)

Anticoagulation reversal should be reserved for patients with ongoing severe bleeding and/or hemodynamic instability, as it may increase the risk of thromboembolic events.<sup>70</sup> Since the publication of the British Society of Gastroenterology, American College of Gastroenterology, and ESGE guidelines,<sup>26,34,37,71</sup> no relevant studies specifically focused on patients with DPPB and vitamin K antagonists (VKAs) have been published to justify statements differing from other sources on LGIB.

Regarding the reversal of direct oral anticoagulants (DOACs), four observational studies have been conducted from 2019 to the present.72-75 Three studies assessed the effectiveness of Idarucizumab in patients on dabigatran therapy,72-74 two of which involved patients with gastrointestinal bleeding.<sup>73,74</sup> The third study did not specify whether gastrointestinal bleeding was included.72 Finally, one sub-analysis of the ANNEXA-IV cohort evaluated the vield of And exanet alpha in patients with gastrointestinal bleeding.75 All studies were observational, lacked a control group, and did not specify if bleeding was a result of prior polypectomy. Therefore, these studies do not provide sufficient evidence to support or refute the recommendation of Idarucizumab or Andexanet alfa. Although no specific evidence supports hematology consultation in this scenario, we recommend it as a precautionary measure due to the high cost, potential risks, and the limited familiarity that other specialists may have with reversal agents.

# Anticoagulation resumption *Statement 18*

SEED recommends resuming anticoagulants (vitamin K antagonists and direct oral anticoagulants) following delayed post-polypectomy bleeding in patients with an indication for anticoagulation. Strong recommendation, low-quality evidence. (Strongly agree: 82.1%; agree: 17.9%)

## Statement 19

SEED suggests resuming anticoagulants following delayed post-polypectomy bleeding on day 7 or earlier. In patients with high thrombotic risk, anticoagulation should preferably be restarted within 72 hours, provided hemostasis has been achieved.

Weak recommendation, very low-quality evidence. (Strongly agree: 57.7%; agree: 34.6%)

Overall, current recommendations suggest resumption within 7 days of the bleeding event, given the overall benefit in reducing thromboembolic complications.<sup>34,37,71</sup> According to the ESGE, it is reasonable to restart warfarin and DOACs from day 7 onwards in patients at low thrombotic risk after interruption. In those with high thrombotic risk, heparin bridging is recommended, preferably within 72 h.<sup>26</sup>

The benefits of resuming anticoagulation after hospitalization for LGIB are well established. In a large cohort study of patients experiencing diverticular hemorrhage, discontinuation of anticoagulation was associated with an increased risk of cardiovascular events, while no significant association was found with rebleeding.<sup>51</sup> Similarly, in a cohort of patients with atrial fibrillation, resumption of VKAs or DOACs following LGIB was linked to a lower hazard of all-cause mortality and a comparable risk of recurrent gastrointestinal bleeding compared to those who did not resume anticoagulation.76 An analysis of patients from the UK audit of LGIB on anti-thrombotics revealed that rebleeding rates in the DOAC (20.0%) and warfarin groups (23.1%) were similar to the unexposed group (19.4%). The use of DOACs or warfarin was not associated with either in-hospital rebleeding or hospital readmission due to further bleeding.<sup>56</sup> Moreover, an analysis of 150 patients with LGIB while on anticoagulants found that resumption of anticoagulation was not associated with recurrent bleeding. When the cohort was expanded to include all patients with GIB, resumption of anticoagulation was linked to a reduced risk of ischemic events during follow-up and decreased mortality.47

On the other hand, a meta-analysis of observational studies demonstrated that resuming anticoagulation after hospitalization for GIB is associated with a reduction in thromboembolic events and mortality, though it may potentially increase the risk of recurrent GIB.<sup>77</sup> Another cohort study found that resumption of anticoagulant therapy was associated with a lower risk of thromboembolism but a higher risk of bleeding, regardless of the timing.<sup>78</sup> The risk of recurrent bleeding was particularly higher in patients with previous bleeding, lower glomerular filtration rate, and those with an index major bleeding event.

# **Bowel preparation**

### Statement 20

SEED suggests the use of oral preparation over enemas or no preparation in the case of suspected delayed post-polypectomy bleeding originating in the right, transverse, or descending colon.

Weak recommendation, low-quality evidence. (Strongly agree: 57.1%; agree: 32.1%)

### Statement 21

SEED suggests that rectosigmoidoscopy may be performed without oral preparation, preferably with enemas, in delayed post-polypectomy bleeding from rectosigmoid lesions.

Weak recommendation; very low-quality evidence. (Strongly agree: 39.3%; agree: 53.9%)

The absence of RCTs and meta-analyses limits the comparison of colon preparation methods for DPPB. Current guidelines support oral preparation for LGIB,<sup>26</sup> which is associated with substantial improvement in the cecal intubation rate.<sup>79,80</sup> Recent data suggest that lower-volume preparation (21) may improve tolerance without compromising the detection of the bleeding site or cecal intubation.<sup>81–83</sup> Without prior preparation, the risk of intestinal perforation is increased.<sup>80</sup> Comparative studies report that polyethylene glycol (PEG) preparation has a superior diagnostic rate of bleeding focus (97.1% vs 84%, p=0.008) and fewer repeat colonoscopies (18.8% vs 44.0%, p<0.001) compared to enemas.<sup>84</sup>

The rationale for omitting bowel preparation is based on the fact that colonoscopy is performed shortly after polypectomy in distal locations, and the laxative effect of the presence of blood may sufficiently clean the colon.<sup>10</sup> Single-center studies explore the role of colonoscopy without preparation in LGIB.<sup>10,79,85,86</sup> Indeed, data exist for patients with DPPB who did not receive preparation before colonoscopy.<sup>87</sup> The cecal intubation rate was 74% in patients prepared with PEG compared to 41% in patients prepared with enema or without preparation, although therapeutic intervention was possible in 15 out of 16 patients with DPPB.<sup>87</sup> Another retrospective analysis included 69 patients with DPPB from polyps <1 cm and no oral preparation. Colonoscopy was successfully performed in all patients with no AEs, and hemostasis was achieved in 90% of patients.<sup>10</sup> The absence of oral preparation has also been proposed for patients admitted to the intensive care unit or elderly patients to avoid associated discomfort.<sup>79,86</sup> Only in cases of single or low-number distal polypectomies (rectum or sigmoid), preparation with enema or without preparation may be considered.<sup>10,84,87</sup>

### Statement 22

SEED recommends using a preparation based on polyethylene glycol (PEG) for colonoscopy following delayed post-polypectomy bleeding. Both high (4 liters) or low volume (2 liters or 1 liter) PEG-based regimens are acceptable.

Strong recommendation, low-quality evidence. (Strongly agree: 53.8%; agree: 38.5%)

Preparations based on PEG 41, and more recently PEG 21 or PEG 11, have been tested according to dosage and timing of the usual preparation intake.81,88 An RCT in patients with LGIB found that low-volume bowel preparation showed the same efficacy and tolerability with better satisfaction compared with high volume,<sup>88</sup> but no direct comparisons of different PEG volumes exist for DPPB patients. Attention should be paid to avoid broncho-aspiration, especially in high-risk situations such as low consciousness levels. We do not recommend administering oral laxatives rapidly, as this increases the risk of AEs. Placement of a nasogastric tube may improve tolerance in patients with limited oral intake and at low risk of aspiration.<sup>26</sup> No evidence exists for sodium picosulfate/magnesium citrate preparations in DPPB.

### Colonoscopy indication Statement 23

SEED suggests against the routine use of colonoscopy for all patients with delayed post-polypectomy bleeding.

Weak recommendation, very low-quality evidence. (Strongly agree: 65.4%; agree: 34.6%)

### Statement 24

SEED suggests performing risk stratification to identify patients with delayed post-polypectomy bleeding who are most likely to benefit from a colonoscopy. Weak recommendation, very low-quality evidence. (Strongly agree: 57.7%; agree: 42.3%)

## Statement 25

In hemodynamically stable patients, SEED suggests an initial observation period (6–24 hours) to assess spontaneous bleeding cessation.

Weak recommendation, very low-quality evidence. (Strongly agree: 57.7%; agree: 42.3%)

## Statement 26

SEED suggests performing a colonoscopy in cases of hourly rectal bleeding or when 3 or more risk factors for active bleeding and/or therapeutic intervention are present (hemodynamic instability at admission, hemoglobin drop  $\geq 2g/dL$ , need for blood transfusion, active antithrombotic use, American Society of Anesthesiologists score  $\geq$  III)

Weak recommendation, very low-quality evidence. (Strongly agree: 42.3%; agree: 50%)

## Statement 27

SEED suggests performing a colonoscopy in patients with persistent bleeding 24 hours after onset, regardless of the presence of risk factors for active bleeding and/ or therapeutic intervention.

Weak recommendation, very low-quality evidence. (Strongly agree: 61.5%; agree: 34.6%)

The decision to perform an invasive procedure is typically based on clinical variables at admission and the physician's expertise. In the systematic literature search conducted for this consensus statement, we did not identify any meta-analysis or RCT evaluating the benefit of colonoscopy in patients with DPBB. Descriptive studies examining the role of colonoscopy are summarized in Supplemental Material. The main finding of these studies, primarily retrospective, is that the rate of repeat colonoscopy ranges from 28% to 100%, although no therapeutic intervention is performed in 0%-63% of cases. Among patients who did not receive an endoscopy and those who underwent colonoscopy without hemostatic intervention, 3%-12% and 0%-20% experienced rebleeding, fewer than 10% were readmitted, and none died during the 30-day follow-up. Therefore, the evidence suggests that a substantial number of patients can be managed without intervention with a low risk of adverse outcomes.<sup>2,7–10,12,18,84,89–92</sup>

A consensus algorithm based on the available evidence and expert opinion for the management of

DPPB is provided in Figure 3. For hemodynamically unstable patients following initial resuscitation, a computed tomography angiography (CTA) scan is preferred, with radiologic embolization if active bleeding is detected. If no active bleeding is identified, a colonoscopy is recommended.

For hemodynamically stable patients, given that bleeding often ceases spontaneously and a significant proportion of patients with DPPB can be managed without a colonoscopy, the spontaneous cessation of bleeding is considered the primary clinical criterion for decision-making. To develop the algorithm, we reviewed the literature to identify factors associated with active bleeding and/or therapeutic intervention (Table 1). Relevant predictors were selected based on their consistency (presence in at least two studies), reproducibility, low interobserver variability, solid pathophysiological basis, and high magnitude of effect (OR>1.5-2.0). The expert panel determined that hourly rectal bleeding is the most significant criterion for guiding decisions due to its indication of high-volume bleeding. The definition of hourly hematochezia is not well established in the literature.<sup>7,9</sup> By consensus, the panel agreed that it should be defined as persistent hematochezia occurring at least once per hour for a period of at least 4h, which clinically indicates significant bleeding. In addition, hemodynamic instability at admission, a hemoglobin drop of  $\geq 2 g/dL$ , need for blood transfusion, antithrombotic use, and American Society of Anesthesiologists score≥III were deemed relevant predictors for the need for colonoscopy.

To date, only two logistic predictive models have been developed. The first model, originating from Australia, was not subjected to internal validation, and no information regarding its calibration or discrimination properties was provided.7 The second model, developed in Spain, underwent internal validation through bootstrapping; however, it has yet to undergo external validation.8 This model provides estimates of the probability of active bleeding. In this study, patients exhibiting three or more risk factors had a predicted probability of >25% of active bleeding, which justified the panel's recommendation for colonoscopy even in the absence of hourly hematochezia, as this subgroup is likely to benefit from the procedure. For hemodynamically stable patients without a high probability of requiring endoscopic

Volume 18



### Figure 3. Proposed management of delayed post-polypectomy bleeding.

\*In case of rectosigmoid lesions consider unprepared rectosigmoidoscopy with enemas. \*\*Persistent hematochezia occurring at least once per hour for a period of at least 4h. \*\*\*Consider colonoscopy, especially in case of ongoing bleeding. ASA, American Society of Anesthesiologists Status Classification; CT, computed tomography; PEG, polyethylene glycol.

Publication	Need for therapeutic intervention	Active bleeding	Moderate or severe bleeding
Rodríguez de Santiago et al. <sup>8</sup>	1. Antiplatelet agents (OR 2.38; 95% CI: 1.35–4.21), $p = 0.003$ 2. Anticoagulant agents (OR 1.76; 95% CI: 1.05–2.97), $p = 0.033$ 3. Hb decrease >2 g/dL (OR 1.63; 95% CI: 1.02–3.12), $p = 0.043$ 4. Hemodynamic instability at admission (OR 3.15; 95% CI: 1.77–5.61), $p < 0.001$ 5. ASA $\ge$ III (OR 2.15; 95% CI: 1.17–3.94), p = 0.014	1. Active use of anticoagulants (OR 2.57; 95% CI: 1.47–4.49), p = 0.001 2. Left-sided polyps (OR 1.95; 95% CI: 1.01–3.75), $p = 0.046$ 3. Prior use of electrocautery (OR 2.60; 95% CI: 1.10–6.10), p = 0.028 4. Pedunculated polyps (OR 1.76; 95% CI: 0.096–3.22), p = 0.069	
Guo et al. <sup>10</sup>	Rebleeding after first re-colonoscopy in DPPB		
Van der Star et al. <sup>9</sup>	1. Hematochezia (≥ hourly) (RR 2.23; <i>p</i> = 0.01)		
Choo and Subhani <sup>132</sup>	Watch and wait strategy useful in the absence of shock index ≥1 and/or a moderate amount of hematochezia more than five occurrences		
Derbyshire et al. <sup>2</sup>	1. Hb drop ≥2 g/dL and/or a blood transfusion (RR 3.47; 95% CI: 1.05–11.52), $p = 0.04$		
Burgess et al. <sup>7</sup>	1. ≥ Hourly hematochezia (OR 36.7; 95% CI: 3.8 to >100), $p = 0.01$ 2. ASA ≥ 2 (OR 20.1; 95% CI: 3.0 to >100), p < 0.001 3.Transfusion (OR 18.7; 95% CI: 2.5 to >100), $p = 0.03$		1. Hemodynamic instability (OR 12.3; 95% CI: 1.04–146), <i>p</i> =0.046 2. Low level of Hb at presentation (OR 0.50; 95% CI: 0.31–0.80), <i>p</i> =0.005

**Table 1.** Predictors of active bleeding, need for therapeutic intervention, or moderate/severe bleeding.

ASA, American Society of Anesthesiologists; CI, confidence interval; DPPB, delayed post-polypectomy bleeding; Hb, hemoglobin; OR, odds ratio; *p*, *p* value; RR, risk ratio.

hemostatic intervention (i.e., <3 risk factors), and given the uncertain calibration of the model in real settings, a 24-h timeframe to assess bleeding cessation appears reasonable.<sup>9</sup> If bleeding persists or recurs, a colonoscopy is recommended.

# Colonoscopy timing

# Statement 28

SEED suggests that for patients requiring colonoscopy, it should be performed at some point during their hospital stay, preferably during the first available non-urgent time slot. The timing of the colonoscopy should be adapted to available resources and the patient's clinical condition.

Weak recommendation, very low-quality evidence. (Strongly agree: 39.3%; agree: 53.6%)

There is no high-quality evidence demonstrating the benefit of early colonoscopy (within the first 24h after admission).<sup>26</sup> We did not find any study showing a difference in clinical outcomes based on the timing of colonoscopy in patients with DPPB. Therefore, the most logical advice seems to follow established recommendations for LGIB. According to current guidelines, patients with major LGIB should undergo a colonoscopy at some point during their hospital stay.<sup>26,34</sup> Subsequent published cohorts are consistent with these recommendations.<sup>93,94</sup>

### Hemostatic therapy

# Indication for hemostatic treatment *Statement 29*

SEED suggests applying an endoscopic hemostatic treatment when there is active bleeding at the postpolypectomy ulcer. Further evidence is needed to clarify the prognostic value and to establish the management of other endoscopic stigmata of recent bleeding.

Weak recommendation, very low-quality evidence. (Strongly agree: 38.5%; agree: 53.8%)

No studies have been conducted to evaluate the benefit of applying a hemostatic endoscopic treatment according to the stigmata found at the postpolypectomy ulcer. Some authors advocate treating only active bleeding, $^{6,95,96}$  while others propose managing additional bleeding stigmata (visible vessel, adhered clot) to reduce the rebleeding rates, similar to Forrest's classification.<sup>7–10,97</sup> Most articles published are retrospective, with several methodological limitations.

The largest study conducted to date evaluates a cohort of 548 patients with DPPB. Among 394 patients managed by colonoscopy, the bleeding point was located in 87.3% of cases: 74 (21.5%) presented with active bleeding, 161 (46.8%) with a visible vessel-adherent clot, and 109 (31.7%) with fibrin-hematin. Endoscopic hemostatic therapy was performed in all lesions with active bleeding, most scars with a visible vesseladherent clot (97.5%), and half of the ulcers with fibrin-hematin. Despite this management, 47 patients rebled within the first 30 days with no differences observed in the rebleeding rate stratified by ulcer stigmata. Rebleeding rates were lower in patients not receiving a colonoscopy (or radiologic/surgical intervention) and also in those who were not treated during re-colonoscopy.8

In another retrospective study (N=69), colonoscopy was performed in 100% of patients, and all but three received endoscopic treatment whether they had active bleeding, visible vessel/adhered clot, or hematin. Greater rebleeding rates were found in patients presenting active bleeding or visible vessel/clot compared with hematin.<sup>10</sup>

Another multicentric study analyzed 42 cases of DPPB. A second colonoscopy was performed in almost three-quarters of patients and hemostatic therapy was applied in 100% of patients with active bleeding (n=6), in 80% with stigmata of recent bleeding (n=16), while individuals with clean ulcer base or not visualized did not receive any therapy. Rebleeding rates were similar among patients receiving endoscopic hemostatic therapy and those left untreated, without significant differences between groups with and without active bleeding.<sup>9</sup>

Ultimately, the ulcer's assessment and the decision to apply a hemostatic approach rely on the endoscopist. Moreover, the interpretation of the culprit ulcer is operator-dependent. In addition, visible vessels and adherent clots are considered into the same category, and visible vessels can be part of a normal post-polypectomy mucosal defect, which complicates the establishment of a stigmata-based therapeutical decision. Applying an endoscopic treatment when there is active bleeding seems logical, but further evidence is needed to clarify the prognostic value and management of visible vessels, adherent clots, pigmented spots, or clean bases.

# First-line and rescue hemostatic treatment **Statement 30**

SEED recommends the use of mechanical therapy as first-line treatment in patients with delayed post-polypectomy bleeding, either as mono- or combined therapy. Thermal and/or injection therapy may also be used, depending on the endoscopist's preference and the devices available.

Strong recommendation, low-quality evidence. (Strongly agree: 69.2%; agree: 30.8%)

## Statement 31

SEED suggests the use of hemostatic powders, mechanical, thermal, and/or injection therapy as rescue treatment, either as mono- or combined therapy, regardless of the approach used during the previous colonoscopy.

Weak recommendation, low-quality evidence. (Strongly agree: 57.7%; agree: 38.5%)

The current scientific evidence evaluating endoscopic treatments for use as first-line or rescue therapy is insufficient to rank these treatments from an evidence-based approach.

In terms of first-line endoscopic treatment in patients with DPPB, the most used devices are through-the-scope clips (TTSCs), which have resulted in very high bleeding arrest rates (between 93% and 100%) and acceptable rebleeding rates (up to 12%).9,11,92,97-108 Although deploying a TTSCs in a post-polypectomy scar can be challenging due to induration, compared to an ulcer immediately after polypectomy, this device is well known and widely available. The largest cohort of patients treated with TTSC alone (N=185)describes an initial hemostasis rate of 100% and an incidence of rebleeding of 10%.<sup>12</sup> Thermal therapy includes both contact modalities (e.g., heater probe, hemostatic forceps, snare-tip coagulation) and non-contact modalities (e.g., argon plasma coagulation). It has been described as an effective monotherapy but carries a risk of transmural injury, so caution is warranted. Sclerosants with or without adrenaline, as well as hemostatic powders, could also be used as first-line treatments.11,92,106,109,110

For combined therapy, sclerosis, thermal therapy, and TTSCs have been used together with good results.<sup>9,97,100,111–113</sup> In the largest DPPB cohort, monotherapy and combined therapy were used in similar proportions, with the most common combination being sclerotherapy and TTSCs.<sup>8</sup>

Rescue treatment modalities have been scarcely reported. TTSCs and over-the-scope clips have shown very high hemostatic rates, with the latter option covering a wider area.<sup>11,99,112</sup> Thermal therapy, hemostatic powders, and band ligation of pedunculated polyps are also viable alternatives for rebleeding.<sup>97,109,111</sup> Each of these therapeutic options can be employed regardless of the approach taken during the previous colonoscopy. Currently, there is no published evidence supporting the use of hemostatic gels in the management of DPPB, further research is needed before incorporating these agents into clinical practice.

## Radiology

### Diagnosis

### Statement 32

SEED suggests performing CT-angiography in patients with delayed post-polypectomy bleeding

experiencing hemodynamic instability or when colonoscopy has failed to control the bleeding, before radiological or surgical treatment.

Weak recommendation, very low-quality evidence. (Strongly agree: 57.7%; agree: 30.8%)

### Statement 33

SEED recommends against the use of other radiological studies (e.g., conventional angiography or red blood cell scintigraphy) due to their limited accuracy in identifying the bleeding origin and their low availability.

Strong recommendation, very low-quality evidence. (Strongly agree: 57.7%; agree: 30.8%)

No studies evaluating the diagnostic accuracy of CTA in the detection of bleeding in patients after an index polypectomy have been conducted. Extrapolating data from other sources of LGIB, CTA has a sensitivity and specificity of 90% and 92%, respectively.<sup>114</sup>

Unlike LGIB secondary to diverticulosis or vascular malformations, where identifying the bleeding origin can be challenging, the location of the DPPB is usually determined by previous colonoscopy. Nevertheless, CT-angiography provides information on the distinction between arterial and venous bleeding and plays a role in the evaluation of the vascular supply prior to selective arterial embolization. Thus, the cecum and ascending and transverse colon are part of the territory of the superior mesenteric artery, while the splenic flexure, descending and sigmoid colon, and rectum correspond to the inferior mesenteric artery. This scheme is generalized, and depending on the patient, there may be variants that make it more complex to establish the vascular supply of the scar.<sup>115</sup> Thus, CTA is useful prior to angiographic embolization when colonoscopy has failed due to massive bleeding, poor bowel preparation, or in hemodynamically unstable patients.2,116

Although conventional angiography could also show the vascular anatomy, it is an invasive procedure requiring higher doses of radiological contrast, is not widely available, and has less sensitivity than CTA.<sup>117</sup> Therefore, angiography is usually performed prior to embolization and after a positive CTA result. Red blood cell scintigraphy has no role in DPPB due to its low availability and greater complexity compared to CTA.

### Treatment

## Statement 34

SEED suggests that angiographic embolization could be of choice in patients with delayed post-polypectomy bleeding either with hemodynamic instability or when colonoscopy has failed to control the bleeding, provided that active bleeding is confirmed by computed tomography angiography.

Weak recommendation, very low-quality evidence. (Strongly agree: 53.8%; agree: 38.5%)

### Statement 35

SEED suggests that angiographic embolization should be elected ahead of surgical treatment in patients with delayed post-polypectomy bleeding either with persistent hemodynamic instability after initial resuscitation or when colonoscopy has failed to control the bleeding Weak recommendation, very low-quality evidence. (Strongly agree: 46.4%; agree: 53.6%)

The evidence evaluating angiographic embolization to control DPPB is limited, with no direct comparisons between radiological intervention and endoscopy or surgery. Considering its greater complexity with the associated risk of colonic ischemia around 1%-4%,<sup>118,119</sup> this technique should be reserved for those patients presenting hemodynamic instability or when colonoscopy has failed. Morbidity and mortality are likely higher in patients with LGIB undergoing surgery,<sup>37</sup> making angiographic embolization preferable to surgery in these cases.

According to the largest cohort of individuals presenting with DPPB, 2.6% of patients required primary angiographic embolization.<sup>8</sup> However, this approach can also be used as rescue therapy in patients experiencing rebleeding.<sup>2,7,8</sup> The technical success rate regarding embolization in patients with DPPB is nearly 100% and the clinical success ranges from 93% to 100%, without major AEs.<sup>2,7,8,120–123</sup> Nonetheless, its success is related to the time elapsed between CTA and arterial embolization, and <90min has been highly associated with detection of extravasation in patients with LGIB.<sup>124</sup>

Endovascular interventions to control bleeding traditionally included intra-arterial vasopressin infusion and mechanical occlusion of the culprit arterial vessel; however, the first option has been replaced by the second one, due to its higher complexity.<sup>125</sup> There are different materials to perform embolization (gelatin sponges,

cyanoacrylate glue, ethylene or polyvinyl alcohol, and microcoils), and the choice will rely on professional experience and local availability. During arterial embolization, it is essential to determine if extravasation is present to reduce the risk of subsequent ischemia, which is related to the inability to achieve super-selective embolization.<sup>116,118,119,126</sup> In addition, despite the absence of extravasated contrast, embolization could be conducted based on CTA findings considering the risk of AEs and the patient's context. Accordingly, TTSCs placed during the previous endoscopy could help with identifying the bleeding location and the vessel supply, even if no extravasation is observed.

### Surgery

### Statement 36

SEED suggests that surgery should only be undertaken in patients with refractory delayed post-polypectomy bleeding when endoscopic or radiological treatment has failed, and always based on local expertise and case-by-case assessment.

Strong recommendation, very low-quality evidence. (Strongly agree: 84.6%; agree: 15.4%)

The proportion of patients experiencing DPPB requiring surgery varies widely, ranging from 0.2% to  $10\%.^{2,7,8,18,127-134}$  This variability in incidence can be attributed to heterogeneity in study designs, differences in the definition of DPPB (where immediate and delayed bleeding are not always distinguished), and timeframes ranging from 14 to 30 days post-procedure.

Up to now, neither RCTs nor non-randomized interventional studies have directly assessed the effectiveness of surgery as a therapeutic option compared to radiological, endoscopic, or conservative approaches. Proposing an RCT to compare the efficacy of surgery with more conservative therapeutic approaches seems illogical and ethically questionable. Given the widespread availability of less invasive alternatives like endoscopic and endovascular radiological interventions and the higher mortality associated with surgery, it is rational to reserve surgery for cases where less aggressive options have failed.

Regarding the surgical approach, segmental colectomy is the most commonly used technique, due to its lower morbidity and mortality compared to extended colectomy.<sup>135</sup> However, in

patients experiencing massive bleeding with multiple post-polypectomy ulcers in whom identifying the responsible one proves challenging after imaging tests or even after an on-table colonoscopy in the operating room, subtotal colectomy seems to be a reasonable option.<sup>130</sup> Considering recent surgical advances of the last few years, laparoscopy emerges as the preferred surgical approach over laparotomy, contingent upon individual experience and the patient's medical status. On an alternative note, transanal surgery is a safe and effective option for managing rectal bleeding.<sup>2</sup> This more conservative surgical approach could even be an alternative to endoscopy in rectal ulcers, although it depends on local expertise. Finally, surgery may be warranted in managing AEs arising from the endoscopic and interventional radiological treatment of DPPB, such as endoscopic perforation or ischemia after radiological embolization.18,131

## Conclusion

This consensus statement provides an evidencebased framework for managing DPPB, addressing a critical gap in current guidelines. The recommendations emphasize individualized care, optimizing healthcare resources, and reducing unnecessary colonoscopies. Given that many cases of DPPB are self-limited, a risk-stratified approach can minimize unnecessary procedures while ensuring timely intervention for high-risk patients.

However, these recommendations rely largely on expert consensus due to limited high-quality evidence, and the proposed algorithm has yet to undergo external validation, which may affect its generalizability. In addition, a formal cost analysis is lacking, though reducing unnecessary colonoscopies could improve resource utilization. To validate these recommendations, we are conducting the COLOHEM-II study, a prospective evaluation of the proposed algorithm. Future research should focus on validating risk stratification models, comparing hemostatic therapies, and integrating costeffectiveness analyses. Despite these limitations, this consensus represents a key step toward standardizing DPPB management, providing a structured and practical guide for clinicians.

## Authorship statement

Guarantor of the article: Enrique Rodríguez de Santiago assumes responsibility as the guarantor

of the article and takes full accountability for the integrity of the work as a whole, from its inception to the final published version.

### Declarations

### Ethics approval and consent to participate

This article does not contain any studies with human or animal participants. There are no human participants in this article and informed consent is not required.

*Consent for publication* Not applicable.

## Author contributions

**Enrique Rodríguez de Santiago:** Investigation; Methodology; Project administration; Writing – original draft; Writing – review & editing.

**Sandra Pérez de la Iglesia:** Investigation; Project administration; Writing – original draft, review & editing.

**Diego de Frutos:** Investigation; Writing – original draft.

José Carlos Marín-Gabriel: Investigation; Writing – original draft.

**Carolina Mangas-SanJuan:** Investigation; Writing – original draft.

**Raúl Honrubia López:** Investigation; Writing – original draft.

**Hugo Uchima:** Investigation; Writing – original draft.

Marta Aicart-Ramos: Investigation; Writing – original draft.

**Miguel Ángel Rodríguez Gandía:** Investigation; Writing – original draft.

**Eduardo Valdivielso Cortázar:** Investigation; Writing – original draft.

Felipe Ramos Zabala: Investigation; Writing – original draft.

**Marco Antonio Álvarez:** Investigation; Writing – original draft.

**Marina Solano Sánchez:** Investigation; Writing – original draft.

**Jesús Manuel González Santiago:** Investigation; Writing – original draft. **Eduardo Albéniz:** Investigation; Writing – original draft.

**Gonzalo Hijos-Mallada:** Investigation; Writing – original draft.

**Nerea Castro Quismondo:** Investigation; Writing – original draft.

**Miguel Fraile-López:** Investigation; Writing – original draft.

**David Martínez Ares:** Investigation; Writing – original draft.

**Javier Tejedor-Tejada:** Investigation; Writing – original draft.

**Luis Hernández:** Investigation; Writing – original draft.

**Joan B. Gornals:** Investigation; Writing – original draft.

**Sergi Quintana-Carbo:** Investigation; Writing – original draft.

**Juan Ocaña:** Investigation; Writing – original draft.

**João A. Cunha Neves:** Investigation; Writing – original draft.

**Juan Martínez Martínez:** Investigation; Writing – original draft.

**María López-Cerón Pinilla:** Investigation; Writing – original draft.

**Carlos Dolz Abadía:** Investigation; Writing – original draft.

**María Pellisé:** Investigation; Methodology; Writing – original draft; Writing – review & editing.

Acknowledgements None.

## Funding

The authors disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: The project did not receive any funding. The open-access fee was paid by the Spanish Society of Gastrointestinal Endoscopy.

## Competing interests

Enrique Rodríguez de Santiago: Olympus (educational activities and advisory), Norgine

(educational activities and congresses fee), Erbe (educational activities), Apollo Endosurgery (educational activities), 3-D Matrix (speaker fee and research grant), Adacyte Therapeutics (advisory), Izasa (speaker fee). José Carlos Marín Gabriel: Olympus, Fujifilm v Alfasigma speaker fees. Raúl Honrubia López: Olympus (educational activities), Norgine (educational activities and congresses fee), Casen Recordati (congresses fee). Hugo Uchima: proctor and consultant (Olympus and ERBE). Joan B. Gornals: Fujifilm: speaker fee and research grant, Boston Scientific: consultant/advisory. María Pellisé: Casen Recordati (speaker, educational activity, and congresses fee), Fujifilm (speaker fee, research grant, and advisory), Olympus (speaker fee, educational activities, and advisory), Alfasigma (speaker fees and advisory), Ipsen-Mayoli (speaker fee), and Norgine (educational activity). The other authors declare no conflict of interest pertaining to this article.

### Availability of data and materials

All relevant material is provided as Supplemental Material.

## ORCID iDs

Enrique Rodríguez de Santiago D https://orcid. org/0000-0002-2852-6042			
Sandra Pérez de la Iglesia D https://orcid.org/ 0000-0002-2029-3985			
José Carlos Marín-Gabriel D https://orcid.org/ 0000-0002-4270-8790			
Carolina Mangas-SanJuan Dhttps://orcid. org/0000-0003-2611-1051			
Hugo Uchima D https://orcid.org/0000-0001- 8411-4993			
Marta Aicart-Ramos D https://orcid.org/ 0000-0002-7092-4562			
Eduardo Valdivielso Cortázar D https://orcid. org/0009-0004-5573-8317			
Marina Solano Sánchez D https://orcid.org/ 0000-0002-4539-9882			
Eduardo Albéniz D https://orcid.org/0000-0002-9311-3746			
Gonzalo Hijos-Mallada D https://orcid.org/ 0000-0003-2743-9593			
Javier Tejedor-Tejada D https://orcid.org/0000- 0002-3585-5733			
Joan B. Gornals D https://orcid.org/0000- 0001-8857-3556			

Sergi Quintana-Carbo D https://orcid.org/0009-0009-7591-1451

João A. Cunha Neves (D) https://orcid.org/ 0000-0001-8354-1619

María Pellisé D https://orcid.org/0000-0001-9739-4998

## Supplemental material

Supplemental material for this article is available online.

## References

- Zauber AG, Winawer SJ, O'Brien MJ, et al. Colonoscopic polypectomy and long-term prevention of colorectal-cancer deaths. N Engl J Med 2012; 366: 687–696.
- Derbyshire E, Hungin P, Nickerson C, et al. Post-polypectomy bleeding in the English National Health Service Bowel Cancer Screening Programme. *Endoscopy* 2017; 49: 899–908.
- Albéniz E, Montori S, Rodríguez de Santiago E, et al. Preventing postendoscopic mucosal resection bleeding of large nonpedunculated colorectal lesions. *Am J Gastroenterol* 2022; 117: 1080–1088.
- Jaruvongvanich V, Prasitlumkum N, Assavapongpaiboon B, et al. Risk factors for delayed colonic post-polypectomy bleeding: a systematic review and meta-analysis. *Int J Colorectal Dis* 2017; 32: 1399–1406.
- Bishay K, Meng ZW, Frehlich L, et al. Prophylactic clipping to prevent delayed colonic post-polypectomy bleeding: meta-analysis of randomized and observational studies. *Surg Endosc* 2022; 36: 1251–1262.
- Sonnenberg A. Management of delayed postpolypectomy bleeding: a decision analysis. *Am J Gastroenterol* 2012; 107: 339–342.
- Burgess NG, Williams SJ, Hourigan LF, et al. A management algorithm based on delayed bleeding after wide-field endoscopic mucosal resection of large colonic lesions. *Clin Gastroenterol Hepatol* 2014; 12: 1525–1533.
- Rodríguez de Santiago E, Hernández-Tejero M, Rivero-Sánchez L, et al. Management and outcomes of bleeding within 30 days of colonic polypectomy in a large, real-life, multicenter cohort study. *Clin Gastroenterol Hepatol* 2021; 19: 732–742.e6.

- 9. van der Star S, Moons LMG, Ter Borg F, et al. Management of delayed bleeding after endoscopic mucosal resection of large colorectal polyps: a retrospective multi-center cohort study. *Endosc Int Open* 2020; 8: E1052–E1060.
- 10. Guo X-F, Yu X-A, Hu J-C, et al. Endoscopic management of delayed bleeding after polypectomy of small colorectal polyps: two or more clips may be safe. *Gastroenterol Rep (Oxf)* 2022; 10: goab051.
- Kim DH and Lim SW. Analysis of delayed postpolypectomy bleeding in a colorectal clinic. *J Korean Soc Coloproctol* 2011; 27: 13–16.
- 12. Lee J-M, Kim WS, Kwak MS, et al. Clinical outcome of endoscopic management in delayed postpolypectomy bleeding. *Intest Res* 2017; 15: 221–227.
- 13. Lacroute J, Marcantoni J, Petitot S, et al. The carbon footprint of ambulatory gastrointestinal endoscopy. *Endoscopy* 2023; 55: 918–926.
- Ferlitsch M, Hassan C, Bisschops R, et al. Colorectal polypectomy and endoscopic mucosal resection: European Society of Gastrointestinal Endoscopy (ESGE) Guideline—Update 2024. *Endoscopy* 2024; 56: 516–545.
- Dewidar O, Lotfi T, Langendam MW, et al. Good or best practice statements: proposal for the operationalisation and implementation of GRADE guidance. *BMJ Evid Based Med* 2023; 28: 189–196.
- Jünger S, Payne SA, Brine J, et al. Guidance on Conducting and REporting DElphi Studies (CREDES) in palliative care: recommendations based on a methodological systematic review. *Palliat Med* 2017; 31: 684–706.
- Feagins LA, Nguyen AD, Iqbal R, et al. The prophylactic placement of hemoclips to prevent delayed post-polypectomy bleeding: an unnecessary practice? A case control study. *Dig Dis Sci* 2014; 59: 823–828.
- Sorbi D, Norton I, Conio M, et al. Postpolypectomy lower GI bleeding: descriptive analysis. *Gastrointest Endosc* 2000; 51: 690–696.
- Yousfi M, Gostout CJ, Baron TH, et al. Postpolypectomy lower gastrointestinal bleeding: potential role of aspirin. *Am J Gastroenterol* 2004; 99: 1785–1789.
- Laursen SB, Oakland K, Laine L, et al. ABC score: a new risk score that accurately predicts mortality in acute upper and lower gastrointestinal bleeding: an international multicentre study. *Gut* 2021; 70: 707–716.

Volume 18

- Oakland K, Jairath V, Uberoi R, et al. Derivation and validation of a novel risk score for safe discharge after acute lower gastrointestinal bleeding: a modelling study. *Lancet Gastroenterol Hepatol* 2017; 2: 635–643.
- Aoki T, Nagata N, Shimbo T, et al. Development and validation of a risk scoring system for severe acute lower gastrointestinal bleeding. *Clin Gastroenterol Hepatol* 2016; 14: 1562–1570.e2.
- 23. Strate LL, Orav EJ and Syngal S. Early predictors of severity in acute lower intestinal tract bleeding. *Arch Intern Med* 2003; 163: 838–843.
- 24. Oakland K, Kothiwale S, Forehand T, et al. External validation of the Oakland score to assess safe hospital discharge among adult patients with acute lower gastrointestinal bleeding in the US. *JAMA Netw Open* 2020; 3: e209630.
- 25. Almaghrabi M, Gandhi M, Guizzetti L, et al. Comparison of risk scores for lower gastrointestinal bleeding: a systematic review and meta-analysis. JAMA Netw Open 2022; 5: e2214253.
- Triantafyllou K, Gkolfakis P, Gralnek IM, et al. Diagnosis and management of acute lower gastrointestinal bleeding: European Society of Gastrointestinal Endoscopy (ESGE) Guideline. *Endoscopy* 2021; 53: 850–868.
- Oakland K. Risk stratification in upper and upper and lower GI bleeding: which scores should we use? *Best Pract Res Clin Gastroenterol* 2019; 42–43: 101613.
- Radaelli F, Rocchetto S, Piagnani A, et al. Scoring systems for risk stratification in upper and lower gastrointestinal bleeding. *Best Pract Res Clin Gastroenterol* 2023; 67: 101871.
- Dindo D, Demartines N and Clavien P-A. Classification of surgical complications: a new proposal with evaluation in a cohort of 6336 patients and results of a survey. *Ann Surg* 2004; 240: 205–213.
- Cotton PB, Eisen GM, Aabakken L, et al. A lexicon for endoscopic adverse events: report of an ASGE workshop. *Gastrointest Endosc* 2010; 71: 446–454.
- Nass KJ, Zwager LW, van der Vlugt M, et al. Novel classification for adverse events in GI endoscopy: the AGREE classification. *Gastrointest Endosc* 2022; 95: 1078–1085.e8.
- 32. Crispino F, Merola E, Tasini E, et al. Adverse events in gastrointestinal endoscopy: validation of the AGREE classification in a real-life 5-year setting. *Dig Liver Dis* 2023; 55: 933–937.

- Pedersen L, Sorensen N, Lindorff-Larsen K, et al. Colonoscopy adverse events: are we getting the full picture? *Scand J Gastroenterol* 2020; 55: 979–987.
- 34. Sengupta N, Feuerstein JD, Jairath V, et al. Management of patients with acute lower gastrointestinal bleeding: an updated ACG guideline. Am J Gastroenterol 2023; 118: 208– 231.
- 35. Alali AA, Almadi MA and Barkun AN. Review article: Advances in the management of lower gastrointestinal bleeding. *Aliment Pharmacol Ther* 2024; 59: 632–644.
- Lewis SR, Pritchard MW, Evans DJ, et al. Colloids versus crystalloids for fluid resuscitation in critically ill people. *Cochrane Database Syst Rev* 2018; 8: CD000567.
- Oakland K, Chadwick G, East JE, et al. Diagnosis and management of acute lower gastrointestinal bleeding: guidelines from the British Society of Gastroenterology. *Gut* 2019; 68: 776–789.
- Bendall O, James J, Pawlak KM, et al. Delayed bleeding after endoscopic resection of colorectal polyps: identifying high-risk patients. *Clin Exp Gastroenterol* 2021; 14: 477–492.
- Kherad O, Restellini S, Martel M, et al. Outcomes following restrictive or liberal red blood cell transfusion in patients with lower gastrointestinal bleeding. *Aliment Pharmacol Ther* 2019; 49: 919–925.
- Odutayo A, Desborough MJR, Trivella M, et al. Restrictive versus liberal blood transfusion for gastrointestinal bleeding: a systematic review and meta-analysis of randomised controlled trials. *Lancet Gastroenterol Hepatol* 2017; 2: 354–360.
- Oakland K, Jairath V and Murphy MF. Advances in transfusion medicine: gastrointestinal bleeding. *Transfus Med* 2018; 28: 132–139.
- 42. Desborough MJR, Colman KS, Prick BW, et al. Effect of restrictive versus liberal red cell transfusion strategies on haemostasis: systematic review and meta-analysis. *Thromb Haemost* 2017; 117: 889–898.
- Mark DG, Huang J, Plimier C, et al. Red blood cell transfusions for emergency department patients with gastrointestinal bleeding within an integrated health system. *Am J Emerg Med* 2020; 38: 746–753.
- 44. Donovan K, Stanworth S and Jairath V. The optimal use of blood components in the management of gastrointestinal bleeding.

Best Pract Res Clin Gastroenterol 2019; 42–43: 101600.

- Miilunpohja S, Jyrkkä J, Kärkkäinen JM, et al. Discontinuing low-dose acetylsalicylic acid after gastrointestinal bleeding is associated with increased mortality. *Scand J Gastroenterol* 2022; 57: 618–624.
- 46. Chan FKL, Leung Ki E-L, Wong GLH, et al. Risks of bleeding recurrence and cardiovascular events with continued aspirin use after lower gastrointestinal hemorrhage. *Gastroenterology* 2016; 151: 271–277.
- Sostres C, Marcén B, Laredo V, et al. Risk of rebleeding, vascular events and death after gastrointestinal bleeding in anticoagulant and/or antiplatelet users. *Aliment Pharmacol Ther* 2019; 50: 919–929.
- Gonai T, Toya Y, Kudara N, et al. Risk factors for short-term re-bleeding in patients with colonic diverticular bleeding: a multicenter retrospective study. *Scand J Gastroenterol* 2022; 57: 690–695.
- Xiao B, Ye Z, Cheng R, et al. Optimal antiplatelet therapy for patients after antiplatelet therapy induced gastrointestinal bleeding: timing. *Intern Emerg Med* 2023; 18: 1385–1396.
- 50. Shiratori Y, Ikeya T, Yamamoto K, et al. Mo1730 continuous administration of antithrombotic drugs does not exacerbate acute lower gastrointestinal bleeding: post-hoc analysis of a multicenter randomized controlled trial. *Gastrointest Endosc* 2020; 91: AB470– AB471.
- Dowd J and Sengupta N. 341 Antithrombotic cessation following hospitalization for diverticular hemorrhage is associated with an increased risk of follow up ischemic and thromboembolic events. *Gastroenterology* 2020; 158: S-60.
- 52. Arnett DK, Blumenthal RS, Albert MA, et al. 2019 ACC/AHA guideline on the primary prevention of cardiovascular disease: executive summary: a report of the American College of Cardiology/American Heart Association Task Force on Clinical Practice Guidelines. J Am Coll Cardiol 2019; 74: 1376–1414.
- 53. Eisenberg MJ, Richard PR, Libersan D, et al. Safety of short-term discontinuation of antiplatelet therapy in patients with drug-eluting stents. *Circulation* 2009; 119: 1634–1642.
- Byrne RA, Rossello X, Coughlan JJ, et al. 2023 ESC Guidelines for the management of acute coronary syndromes. *Eur Heart J* 2023; 44: 3720–3826.

- Ma H, Fan X, Jiao L, et al. Time of resumption of antiplatelet drugs after upper gastrointestinal hemorrhage. *Med Sci Monit* 2022; 28: e936953.
- 56. Oakland K, Desborough MJ, Murphy MF, et al. Rebleeding and mortality after lower gastrointestinal bleeding in patients taking antiplatelets or anticoagulants. *Clin Gastroenterol Hepatol* 2019; 17: 1276–1284.e3.
- 57. Stefanescu Schmidt AC, Steg PG, Yeh RW, et al. Interruption of dual antiplatelet therapy within six months after coronary stents (from the dual antiplatelet therapy study). *Am J Cardiol* 2019; 124: 1813–1820.
- 58. Wang H-Y, Xu B, Song C-X, et al. Thrombotic vs. bleeding events of interruption of dual antiplatelet therapy within 12 months among patients with stent-driven high ischemic risk definition following PCI. *J Interv Cardiol* 2022; 2022: 3895205.
- Gralnek IM, Stanley AJ, Morris AJ, et al. Endoscopic diagnosis and management of nonvariceal upper gastrointestinal hemorrhage (NVUGIH): European Society of Gastrointestinal Endoscopy (ESGE) Guideline—Update 2021. Endoscopy 2021; 53: 300–332.
- 60. Zakko L, Rustagi T, Douglas M, et al. No benefit from platelet transfusion for gastrointestinal bleeding in patients taking antiplatelet agents. *Clin Gastroenterol Hepatol* 2017; 15: 46–52.
- 61. Baharoglu MI, Cordonnier C, Al-Shahi Salman R, et al. Platelet transfusion versus standard care after acute stroke due to spontaneous cerebral haemorrhage associated with antiplatelet therapy (PATCH): a randomised, open-label, phase 3 trial. *Lancet* 2016; 387: 2605–2613.
- 62. Baharoglu MI, Al-Shahi Salman R, Cordonnier C, et al. PATCH trial: explanatory analyses. *Blood* 2020; 135: 1406–1409.
- 63. Oakland K, Guy R, Uberoi R, et al. Acute lower GI bleeding in the UK: patient characteristics, interventions and outcomes in the first nationwide audit. *Gut* 2018; 67: 654–662.
- 64. Majeed A, Hwang H-G, Eikelboom JW, et al. Effectiveness and outcome of management strategies for dabigatran- or warfarin-related major bleeding events. *Thromb Res* 2016; 140: 81–88.
- 65. Garcia DA, Fisher DA, Mulder H, et al. Gastrointestinal bleeding in patients with atrial fibrillation treated with Apixaban or warfarin: insights from the Apixaban for Reduction in Stroke and Other Thromboembolic Events in Atrial Fibrillation (ARISTOTLE) trial. *Am Heart J* 2020; 221: 1–8.

- 66. Held C, Hylek EM, Alexander JH, et al. Clinical outcomes and management associated with major bleeding in patients with atrial fibrillation treated with apixaban or warfarin: insights from the ARISTOTLE trial. *Eur Heart J* 2015; 36: 1264–1272.
- Ono S, Ishikawa M, Matsuda K, et al. Clinical impact of the perioperative management of oral anticoagulants in bleeding after colonic endoscopic mucosal resection. *BMC Gastroenterol* 2019; 19: 206.
- 68. Rodríguez de Santiago E, Sánchez Aldehuelo R, Riu Pons F, et al. Endoscopy-related bleeding and thromboembolic events in patients on direct oral anticoagulants or vitamin K antagonists. *Clin Gastroenterol Hepatol* 2022; 20: e380–e397.
- 69. Tomaselli GF, Mahaffey KW, Cuker A, et al. 2020 ACC Expert Consensus Decision Pathway on management of bleeding in patients on oral anticoagulants: a report of the American College of Cardiology Solution Set Oversight Committee. *J Am Coll Cardiol* 2020; 76: 594–622.
- Martin A-C, Benamouzig R, Gouin-Thibault I, et al. Management of gastrointestinal bleeding and resumption of oral anticoagulant therapy in patients with atrial fibrillation: a multidisciplinary discussion. Am J Cardiovasc Drugs 2023; 23: 407–418.
- Veitch AM, Radaelli F, Alikhan R, et al. Endoscopy in patients on antiplatelet or anticoagulant therapy: British Society of Gastroenterology (BSG) and European Society of Gastrointestinal Endoscopy (ESGE) guideline update. *Endoscopy* 2021; 53: 947–969.
- 72. Crespo-Leiro MG, López-Vilella R, López Granados A, et al. Use of Idarucizumab to reverse the anticoagulant effect of dabigatran in cardiac transplant surgery: a multicentric experience in Spain. *Clin Transplant* 2019; 33: e13748.
- 73. Van der Wall SJ, Lopes RD, Aisenberg J, et al. Idarucizumab for dabigatran reversal in the management of patients with gastrointestinal bleeding. *Circulation* 2019; 139: 748–756.
- 74. van der Wall SJ, van Rein N, van den Bemt B, et al. Performance of idarucizumab as antidote of dabigatran in daily clinical practice. *Europace* 2019; 21: 414–420.
- 75. Siegal DM, Forbes N, Eikelboom J, et al. The efficacy and safety of andexanet alfa in patients with acute gastrointestinal bleeding while taking factor Xa inhibitors: an ANNEXA-4 subanalysis. *Circulation* 2024; 149: 1315–1318.
- 76. Rajan D, Bonde AN, Olesen JB, et al. P1256 Resumption of anticoagulant treatment in

patients with atrial fibrillation following gastrointestinal bleeding: a nationwide cohort study. *Eur Heart*  $\mathcal{J}$  2019; 40: ehz748.0214.

- 77. Tapaskar N, Pang A, Werner DA, et al. Resuming anticoagulation following hospitalization for gastrointestinal bleeding is associated with reduced thromboembolic events and improved mortality: results from a systematic review and meta-analysis. *Dig Dis Sci* 2021; 66: 554–566.
- 78. Candeloro M, van Es N, Cantor N, et al. Recurrent bleeding and thrombotic events after resumption of oral anticoagulants following gastrointestinal bleeding: communication from the ISTH SSC Subcommittee on Control of Anticoagulation. *J Thromb Haemost* 2021; 19: 2618–2628.
- 79. Repaka A, Atkinson MR, Faulx AL, et al. Immediate unprepared hydroflush colonoscopy for severe lower GI bleeding: a feasibility study. *Gastrointest Endosc* 2012; 76: 367–373.
- Jensen DM, Machicado GA, Jutabha R, et al. Urgent colonoscopy for the diagnosis and treatment of severe diverticular hemorrhage. N Engl J Med 2000; 342: 78–82.
- Pontone S, Palma R, Panetta C, et al. Polyethylene glycol-based bowel preparation before colonoscopy for selected inpatients: a pilot study. J Dig Dis 2018; 19: 40–47.
- Yang D, Summerlee R, Rajca B, et al. A pilot study to evaluate the feasibility of implementing a split-dose bowel preparation for inpatient colonoscopy: a single-center experience. *BMJ Open Gastroenterol* 2014; 1: e000006.
- Yadlapati R, Johnston ER, Gluskin AB, et al. An automated inpatient split-dose bowel preparation system improves colonoscopy quality and reduces repeat procedures. *J Clin Gastroenterol* 2018; 52: 709–714.
- Lim DS, Kim HG, Jeon SR, et al. Comparison of clinical effectiveness of the emergent colonoscopy in patients with hematochezia according to the type of bowel preparation. *J Gastroenterol Hepatol* 2013; 28: 1733–1737.
- 85. Correia C, Almeida N, Souto P, et al. Is urgent colonoscopy without bowel preparation really useful? Colonoscopy without bowel preparation. *Rev Esp Enferm Dig* 2022; 114: 632.
- Utku ÖG and Karatay E. Immediate unprepared polyethylene glycol-flush colonoscopy in elderly patients with severe lower gastrointestinal bleeding. *Geriatr Gerontol Int* 2020; 20: 559–563.
- Saito K, Inamori M, Sekino Y, et al. Management of acute lower intestinal bleeding:

what bowel preparation should be required for urgent colonoscopy? *Hepatogastroenterology* 2009; 56: 1331–1334.

- Saviano A, Petruzziello C, Riccioni ME, et al. Lower gastrointestinal bleeding in the emergency department: high-volume vs. lowvolume peg bowel preparation for colonoscopy: a randomized trial. *Rev Recent Clin Trials* 2023; 18: 76–81.
- Chiba H, Ohata K, Tachikawa J, et al. Delayed bleeding after colorectal endoscopic submucosal dissection: when is emergency colonoscopy needed? *Dig Dis Sci* 2019; 64: 880–887.
- 90. Hayasaka J, Yamashita S, Matsui A, et al. Safety of cold snare polypectomy during continuous use of antithrombotic drugs for delayed postpolypectomy bleeding: a pilot study. *Dig Dis* 2023; 41: 729–736.
- 91. Kishino T, Oyama T, Hotta K, et al. Risk of colonoscopic post-polypectomy bleeding in patients after the discontinuation of antithrombotic therapy. *Turk J Gastroenterol* 2020; 31: 752–759.
- 92. Feagins LA, Smith AD, Kim D, et al. Efficacy of prophylactic hemoclips in prevention of delayed post-polypectomy bleeding in patients with large colonic polyps. *Gastroenterology* 2019; 157: 967–976.e1.
- Özkan A. Timing of colonoscopy for hemodynamically stable patients with acute lower gastrointestinal bleeding. Surg Laparosc Endosc Percutan Tech 2023; 33: 76–78.
- 94. Lahat A, Klang E, Rahman N, et al. Utility of early colonoscopy for acute lower gastrointestinal bleeding: a retrospective cohort study. *Therap Adv Gastroenterol* 2023; 16: 17562848221147757.
- 95. Ma MX and Bourke MJ. Complications of endoscopic polypectomy, endoscopic mucosal resection and endoscopic submucosal dissection in the colon. *Best Pract Res Clin Gastroenterol* 2016; 30: 749–767.
- Park S-K, Seo JY, Lee M-G, et al. Prospective analysis of delayed colorectal post-polypectomy bleeding. *Surg Endosc* 2018; 32: 3282–3289.
- Parra-Blanco A, Kaminaga N, Kojima T, et al. Hemoclipping for postpolypectomy and postbiopsy colonic bleeding. *Gastrointest Endosc* 2000; 51: 37–41
- Averbach M, D'Assunção MA, Costa RD, et al. The use of two endoscopes for treatment of postpolypectomy hemorrhage. *Gastrointest Endosc* 2006; 63: 1084–1086.

- Binmoeller KF, Thonke F and Soehendra N. Endoscopic hemoclip treatment for gastrointestinal bleeding. *Endoscopy* 1993; 25: 167–170.
- 100. Chang K, Lee BS, Tekeste T, et al. The effect of prophylactic hemoclips on the risk of delayed post-endoscopic mucosal resection bleed for upper and lower gastrointestinal lesions: a retrospective cohort study. *BMC Gastroenterol* 2020; 20: 60.
- 101. Fujishiro M, Yahagi N, Nakamura M, et al. Successful outcomes of a novel endoscopic treatment for GI tumors: endoscopic submucosal dissection with a mixture of highmolecular-weight hyaluronic acid, glycerin, and sugar. *Gastrointest Endosc* 2006; 63: 243–249.
- 102. Ji J-S, Lee S-W, Kim TH, et al. Comparison of prophylactic clip and endoloop application for the prevention of postpolypectomy bleeding in pedunculated colonic polyps: a prospective, randomized, multicenter study. *Endoscopy* 2014; 46: 598–604.
- 103. Lee BG, Shin SH, Lee YA, et al. Juvenile polyp and colonoscopic polypectomy in childhood. *Pediatr Gastroenterol Hepatol Nutr* 2012; 15: 250–255.
- 104. Lee CK, Lee S-H, Park J-Y, et al. Prophylactic argon plasma coagulation ablation does not decrease delayed postpolypectomy bleeding. *Gastrointest Endosc* 2009; 70: 353–361.
- 105. Mashimo Y, Matsuda T, Uraoka T, et al. Endoscopic submucosal resection with a ligation device is an effective and safe treatment for carcinoid tumors in the lower rectum. *β Gastroenterol Hepatol* 2008; 23: 218–221.
- 106. Motz VL, Lester C, Moyer MT, et al. Hybrid argon plasma coagulation-assisted endoscopic mucosal resection for large sessile colon polyps to reduce local recurrence: a prospective pilot study. *Endoscopy* 2022; 54: 580–584.
- 107. Shioji K, Suzuki Y, Kobayashi M, et al. Prophylactic clip application does not decrease delayed bleeding after colonoscopic polypectomy. *Gastrointest Endosc* 2003; 57: 691–694.
- 108. Soh JS, Seo M and Kim K-J. Prophylactic clip application for large pedunculated polyps before snare polypectomy may decrease immediate postpolypectomy bleeding. *BMC Gastroenterol* 2020; 20: 68.
- 109. Hookey L, Barkun A, Sultanian R, et al. Successful hemostasis of active lower GI

bleeding using a hemostatic powder as monotherapy, combination therapy, or rescue therapy. *Gastrointest Endosc* 2019; 89: 865– 871.

- Rodríguez de Santiago E, Burgos-Santamaría D, Pérez-Carazo L, et al. Hemostatic spray powder TC-325 for GI bleeding in a nationwide study: survival and predictors of failure via competing risks analysis. *Gastrointest Endosc* 2019; 90: 581–590.e6.
- Slivka A, Parsons WG and Carr-Locke DL. Endoscopic band ligation for treatment of postpolypectomy hemorrhage. *Gastrointest Endosc* 1994; 40: 230–232.
- Armellini E, Crinò SF, Orsello M, et al. Novel endoscopic over-the-scope clip system. World J Gastroenterol 2015; 21: 13587–13592.
- 113. Pioche M, Camus M, Rivory J, et al. A selfassembling matrix-forming gel can be easily and safely applied to prevent delayed bleeding after endoscopic resections. *Endosc Int Open* 2016; 4: E415–E419.
- He B, Yang J, Xiao J, et al. Diagnosis of lower gastrointestinal bleeding by multi-slice CT angiography: a meta-analysis. *Eur J Radiol* 2017; 93: 40–45.
- 115. Miller M and Smith TP. Angiographic diagnosis and endovascular management of nonvariceal gastrointestinal hemorrhage. *Gastroenterol Clin North Am* 2005; 34: 735–752.
- 116. Weldon DT, Burke SJ, Sun S, et al. Interventional management of lower gastrointestinal bleeding. *Eur Radiol* 2008; 18: 857–867.
- 117. Hastings GS. Angiographic localization and transcatheter treatment of gastrointestinal bleeding. *Radiographics* 2000; 20: 1160–1168.
- 118. Yata S, Ihaya T, Kaminou T, et al. Transcatheter arterial embolization of acute arterial bleeding in the upper and lower gastrointestinal tract with *N*-butyl-2-cyanoacrylate. J Vasc Interv Radiol 2013; 24: 422–431.
- 119. Ali M, Ul Haq T, Salam B, et al. Treatment of nonvariceal gastrointestinal hemorrhage by transcatheter embolization. *Radiol Res Pract* 2013; 2013: 604328.
- 120. d'Othée BJ, Surapaneni P, Rabkin D, et al. Microcoil embolization for acute lower gastrointestinal bleeding. *Cardiovasc Intervent Radiol* 2006; 29: 49–58.

- 121. Koganemaru M, Abe T, Iwamoto R, et al. Ultraselective arterial embolization of vasa recta using 1.7-French microcatheter with small-sized detachable coils in acute colonic hemorrhage after failed endoscopic treatment. AJR Am J Roentgenol 2012; 198: W370-W372.
- 122. Gravante G, Delogu D and Venditti D. Postendoscopic polypectomy delayed bleeding concomitant with an abdominoperineal resection: a case report. *Eur Rev Med Pharmacol Sci* 2007; 11: 355–357.
- 123. Frodsham A, Berkmen T, Ananian C, et al. Initial experience using N-butyl cyanoacrylate for embolization of lower gastrointestinal hemorrhage. J Vasc Interv Radiol 2009; 20: 1312–1319.
- 124. Koh FH, Soong J, Lieske B, et al. Does the timing of an invasive mesenteric angiography following a positive CT mesenteric angiography make a difference? *Int J Colorectal Dis* 2015; 30: 57–61.
- 125. Loffroy R, Falvo N, Nakai M, et al. When all else fails—radiological management of severe gastrointestinal bleeding. *Best Pract Res Clin Gastroenterol* 2019; 42–43: 101612.
- 126. Lenhart M, Paetzel C, Sackmann M, et al. Superselective arterial embolisation with a liquid polyvinyl alcohol copolymer in patients with acute gastrointestinal haemorrhage. *Eur Radiol* 2010; 20: 1994–1999.
- 127. Buddingh KT, Herngreen T, Haringsma J, et al. Location in the right hemi-colon is an independent risk factor for delayed postpolypectomy hemorrhage: a multi-center casecontrol study. *Am J Gastroenterol* 2011; 106: 1119–1124.
- 128. Gupta S, Sidhu M, Shahidi N, et al. Effect of prophylactic endoscopic clip placement on clinically significant post-endoscopic mucosal resection bleeding in the right colon: a singlecentre, randomised controlled trial. *Lancet Gastroenterol Hepatol* 2022; 7: 152–160.
- 129. Witt DM, Delate T, McCool KH, et al. Incidence and predictors of bleeding or thrombosis after polypectomy in patients receiving and not receiving anticoagulation therapy. *J Thromb Haemost* 2009; 7: 1982– 1989.
- Plummer JM, Gibson TN, Mitchell DIG, et al. Emergency subtotal colectomy for lower gastrointestinal haemorrhage: over-utilised or

## E Rodríguez de Santiago, S Pérez de la Iglesia et al.

under-estimated? Int J Clin Pract 2009; 63: 865–868.

- Sawhney MS, Salfiti N, Nelson DB, et al. Risk factors for severe delayed postpolypectomy bleeding. *Endoscopy* 2008; 40: 115–119.
- 132. Choo WK and Subhani J. Complication rates of colonic polypectomy in relation to polyp characteristics and techniques: a district hospital experience. J Interv Gastroenterol 2012; 2: 8–11.
- 133. Pigò F, Bertani H, Manno M, et al. Colonic postpolypectomy bleeding is related to polyp size and heparin use. *Clin Endosc* 2017; 50: 287–292.
- 134. Rosen L, Bub DS, Reed JF, et al. Hemorrhage following colonoscopic polypectomy. *Dis Colon Rectum* 1993; 36: 1126–1131.
- 135. You YN, Chua HK, Nelson H, et al. Segmental vs. extended colectomy: measurable differences in morbidity, function, and quality of life. *Dis Colon Rectum* 2008; 51: 1036–1043.

Visit Sage journals online journals.sagepub.com/ home/tag

Sage journals